Author's response to reviews

Title: Measurement of matrix metalloproteinase 9-mediated Collagen type III degradation fragment as a marker of skin fibrosis.

Authors:

Efstathios Vassiliadis (eva@nordicbioscience.com)
Sanne Skovgard Veidal (ssv@nordicbioscience.com)
Natasha Barascuk (nb@nordicbioscience.com)
Jhinuk Basu Mullick (jbm@nordicbioscience.com)
Rikke Elgaard Clausen (rec@nordicbioscience.com)
Lise Larsen (ll@nordicbioscience.com)
Henrik Simonsen (hsi@nordicbioscience.com)
Dorthe Vang Larsen (dvl@nordicbioscience.com)
Anne-Christine Bay Jensen (acb@nordicbioscience.com)
Toni Segovia Silvestre (as@nordicbioscience.com)
Diana Julie Leeming (djl@nordicbioscience.com)
Morten A Karsdal (mk@nordicbioscience.com)

Version: 9 Date: 21 March 2011

Author's response to reviews:

All suggested revisions were carried out as suggested

Abstract: abstracts should not cite references, nor refer to figures or tables-REFERENCES WERE EXCLUDED

Abstract: please ensure that the Abstract present in the manuscript matches exactly what is in the submission system - UPDATED ABSTRACT WAS UPLOADED

Authors' contributions: Please use initials to refer to each author - WAS CHANGED TO INITIALS

Authors' contributions: Please indicate that all authors read and approved the final manuscript - IT IS NOW STATED IN THE RELEVANT SECTION

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance- IT WAS CHECKED AGAIN AS SUGGESTED